Author's response to reviews

Title: Somatic VHL gene alterations in MEN2-associated medullary thyroid carcinoma

Authors:

Christian A Koch (ckoch@medicine.umsmed.edu)
Frederieke M Brouwers (brouwerf@mail.nih.gov)
Alexander O Vortmeyer (vortmeyera@ninds.nih.gov)
Andrea Tannapfel (andrea.tannapfel@rub.de)
Steven K Libutti (libuttis@mail.nih.gov)
Zhengping Zhuang (zhuangz@ninds.nih.gov)
Karel Pacak (karel@mail.nih.gov)
Hartmut P Neumann (neumann@Medizin.Ukl.Uni-Freiburg.De)
Ralf Paschke (ralf.paschke@medizin.uni-leipzig.de)

Version: 2 Date: 15 March 2006

Author's response to reviews:

References have been numbered and put into the style for Biomed Central. Abstract has been structured into requested style. Competing interests, abbreviations, authors contributions have been added.